Author's response to reviews

Title: Conservative treatment of idiopathic scoliosis according to FITS concept: presentation of the method and preliminary, short term radiological and clinical results based on SOSORT and SRS criteria.

Authors:

Marianna PT, PhD Bialek MB (mbfits@o2.pl)

Version: 5 Date: 7 November 2011

Author's response to reviews: see over
Concept FITS - Functional Individual Therapy Scoliosis - is a new Polish method of treatment of idiopathic scoliosis presented at many national conferences as well as foreign - SOSORT. The author presents the first this concept based on early results of statistical surveys.